Author's response to reviews

Title: Dietary intake in 6-year-old children from southern Poland: part 2 - vitamin and mineral intakes

Authors:

Sylwia Merkiel (sylwiamerkiel@awf.poznan.pl)
Wojciech Chalcarz (chalcarz@awf.poznan.pl)

Version: 3
Date: 3 December 2014

Author's response to reviews: see over
Dear Doctor Peter O'Donovan,

We send our revised manuscript titled: 'Dietary intake in 6-year-old children from southern Poland: part 2 – vitamin and mineral intakes' which we kindly ask to be published in the BMC Pediatrics.

We are very glad and honoured that our manuscript has been assessed as suitable for publication in the BMC Pediatrics. Thank you very much.

We introduced the section 'Acknowledgements' as requested and we moved to this section the information about the source of funding. The funds which we received from the Polish Ministry of Science and Higher Education were used to carry out the research and it was the only role of the funding body. We did not add the name of a language editor because we revised the manuscript by ourselves – we had been analysing the whole manuscript very carefully to improve the language and that is why we used the whole time provided for the revision.

We hope that our manuscript will be accepted for publication.

We are looking forward to receiving your decision.

Yours sincerely,
Sylwia Merkiel and Wojciech Chalcarz